Author's response to reviews

Title: A prospective study of health care resource utilisation and costs associated with schizophrenia in France

Authors:

Emmanuelle Sarlon (emmanuelle.Sarlon@ch-senlis.fr)
Aurélie Millier (ami@creativ-ceutical.com)
Dirk Heider (heidd@medizin.uni-leipzig.de)
Jean-Michel Azorin (jean-michel.azorin@mail.ap-hm.fr)
Hans-Helmut König (hans-helmut.koenig@medizin.uni-leipzig.de)
Karina Hansen (KHAN@Lundbeck.com)
Matthias Angermeyer (angermeyer@aon.at)
Samuel Aballéa (sab@creativ-ceutical.com)
Mondher Toumi (mondher.toumi@univ-lyon1.fr)

Version: 4 Date: 5 July 2012

Author's response to reviews: see over
1 Reviewer: Roshni Mangalore

Minor correction

1. The last sentence in section 2.2 is a repeat from the previous section. Please deal with this.

This has been addressed. Thank you for noticing it.

Level of interest: An article of importance in its field

Quality of written English: Acceptable

Statistical review: Yes, and I have assessed the statistics in my report.

Declaration of competing interests:
'I declare that I have no competing interests'

2 Reviewer: David Sclar

The authors have addressed the requisite issues. The manuscript is now acceptable for BMC HSR.